Mediating Factors Associated With Physical Activity in Older Adults at Increased Dementia **Risk**

American Journal of Alzheimer's Disease & Other Dementias® Volume 39: 1-15 © The Author(s) 2024 Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/15333175241257849 journals.sagepub.com/home/aja



Maria Isabel Cardona, MA¹, Melanie Luppa, PD, Dr. rer. med², Andrea Zülke, Dr. rer. nat², Eric S. Kroeber, Dr. med³, Alexander Bauer, Dr³, Juliane Döhring, Dr. med⁴, Catharina Escales, Dr. med⁴, Christian Brettschneider, PD, Dr⁵, Thomas Frese, Dr. med³, Robert P. Kosilek, Dr.⁶, Jochen Gensichen, Dr. med⁶, Wolfgang Hoffmann, Dr. med, MPH^{1,8}, Hanna Kaduszkiewicz, Dr. med⁴, Hans-Helmut König, Dr.⁵, Birgitt Wiese, Dr.⁷, Steffi G. Riedel-Heller, Dr. med², and

Jochen René Thyrian, Dr. rer. med^{1,8}

Abstract

While regular physical-activity (PA) is beneficial, multimorbid individuals at increased dementia risk may exhibit reduced PA levels. Thus, a more comprehensive understanding of mediating factors responsible for inactivity in this population is needed. This study investigated the impact of a multimodal intervention on PA changes at 24-month follow-up and associated mediating factors among community-dwelling patients aged 60-77, with increased dementia risk determined by the CAIDE Dementia Risk Score. Of 1030 participants recruited, 819 completed the assessment. Thus, a generalized estimating equations model initially assessed differences in PA over 24 months, followed by a tree analysis identifying mediating factors influencing PA changes postintervention. While no significant effect on regular PA was found during the follow-up (P = .674), subgroup analysis revealed improved self-efficacy (P = .000) associated with increased engagement in PA. Incorporating self-efficacy elements into future strategies is crucial for promoting PA among individuals with multimorbidity and at increased dementia risk.

Keywords

mediating factors, physical activity, multimorbidity, dementia, decision tree analysis

¹German Center for Neurodegenerative Diseases (DZNE), Site Rostock/Greifswald, Greifswald, Germany

²Institute of Social Medicine, Occupational Health and Public Health (ISAP), Medical Faculty, University of Leipzig, Leipzig, Germany

³Institute of General Practice and Family Medicine, Martin-Luther-University Halle-Wittenberg, Halle, Germany

⁴Institute of General Practice, University of Kiel, Kiel, Germany

⁵Department of Health Economics and Health Service Research, University Medical Center Hamburg-Eppendorf, Hamburg, Germany

⁶Institute of General Practice and Family Medicine, University Hospital, LMU Munich, Munich, Germany

⁷MHH Information Technology, Hannover Medical School, Hannover, Germany

⁸Institute for Community Medicine, University Medicine Greifswald, Greifswald, Germany

Corresponding Author:

Jochen René Thyrian, German Center for Neurodegenerative Diseases (DZNE), Site Rostock/Greifswald, Ellernholzstraße 1, Greifswald 17489, Germany. Email: Rene.Thyrian@dzne.de



Creative Commons Non Commercial No Derivs CC BY-NC-ND: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs 4.0 License (https://creativecommons.org/licenses/by-nc-nd/4.0/) which permits non-commercial use, reproduction and distribution of the work as published without adaptation or alteration, without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

Data Availability Statement included at the end of the article

Significance Statement

This manuscript uses a unique methodology to study factors contributing to inactivity in older adults with multimorbidity and at increased dementia risk. Our approach identifies specific mediating factors including self-efficacy and subgroups within the population with shared characteristics related to desired health behaviour (physical activity). This information can help target interventions more effectively, improving physical activity promotion, guiding policies, and resource allocation in public health.

Introduction

As the world's population ages, the prevalence of dementia is increasing.¹ It is estimated that there will be a global increment of 57.4 million cases of dementia in 2019 to 152.8 million cases in 2050.² Indeed, the growth in the number of persons living with dementia represents a primary worldwide challenge because of its physical, psychological and economic repercussions.³ Together with its global prevalence and impact, the absence of a cure and treatment for dementia highlights the relevance of prevention as an effective way to reduce dementia implications.¹ Therefore, currently, both scientific research and global health strategies have focused on dementia prevention efforts by addressing modifiable risk factors, including physical inactivity.^{4,5} Approximately 40% of dementia cases worldwide can be linked to modifiable risk factors.⁶ Likewise, multimorbidity has a robust association with mild cognitive impairment and subsequent dementia.⁷ Particularly, older adults with cardiovascular, sensory impairment and neuropsychiatric multimorbidity are affected.⁸ Thus, by recognizing and targeting modifiable and associated risk factors for dementia, key elements are highlighted for primary prevention strategies and dementia incidence reduction.

Multimodal interventions that target modifiable risk factors show promise in preventing cognitive decline. Notable European studies such as The Finnish Geriatric Intervention Study to Prevent Cognitive Impairment and Disability (FINGER),⁹ the French Multimodal Alzheimer Preventive Trial (MAPT),¹⁰ and Dutch Prevention of Dementia by Intensive Vascular Care (PreDIVA)¹¹ have demonstrated the efficacy of such approaches. The FINGER Trial,⁹ lasting two years, displayed significant cognitive improvements among at-risk older adults through lifestyle interventions. Similarly, the 3-year MAPT Trial¹⁰ displayed positive cognitive effects despite not meeting its primary goal. Lastly, the 6-year Pre-DIVA Study¹¹ revealed a reduction in dementia incidence among individuals with untreated hypertension who adhered to treatment. These trials highlight the potential of targeting modifiable risk factors to reduce dementia risk.

Within these interventions, physical activity (PA) is consistently included as a crucial component, with the aim of mitigating various dementia risk factors, promoting brain health and enhancing overall well-being.⁴ Long-term engagement in PA has been linked to cognitive benefits, improved physical functioning, and mental health, reducing dementia risk and delaying dementia progression.^{12,13} Moreover, a strong correlation exists between PA and reduced non-communicable diseases (NCDs) and decreased mortality risk linked to multimorbidity.¹⁴ In order to achieve those health benefits, the World Health Organization (WHO) recommends a minimum of 150 min/per week of moderateintensity aerobic PA for older adults.¹⁵

Despite PA benefits, the effectiveness of PA in delaying cognitive decline is not yet fully established,⁴ especially considering the challenges of engaging and maintaining adherence among older adults.^{16,17} Evidence shows that in Germany, less than 150 minutes of PA per week is reported by 65% of individuals aged 65 or older, indicating a shortfall in meeting WHO recommendation.¹⁸ Similarly, adherence to WHO PA recommendations was found to be less than a third among older adults with multimorbidity.¹⁹ Such a considerable absence of adherence to recommended PA may contribute to worsening multimorbidity or other adverse health outcomes.¹⁹

Given the extensive incidence of low PA levels in older adults with multimorbidity and increased dementia risk, understanding the mediating factors influencing PA engagement within multimodal interventions is essential. Additionally, investigating the impact of multimodal interventions on PA changes can provide valuable insights into their effectiveness.²⁰ Potential mediating factors influencing engagement in PA interventions among older adults with multimorbidity and at increased dementia risk, as proposed by the socioecological model, encompass both individual and environmental factors.^{21,22} Notably, recurring barriers often hinder PA participation among people with multimorbidity, including physical barriers (e.g. pain and fatigue), psychological constraints (depression, anxiety, fear of falling, extrinsic motivation, feelings of isolation, knowledge about the condition), and financial limitations.²⁰ While previous literature provide some evidence, regarding mediating factors for regular PA in persons with multimorbidity, prior research presents a limitation by studying PA mediating factors without considering possible high-risk subgroups.²³ Decision tree analysis plays an essential role in public health research by analysing the intricate interactions across outcomes and mediating factors, identifying high-risk subgroups to implement preventive measures and interventions.²⁴ Thus, decision trees enable the identification of distinct subgroups within the population that have common and distinguishable characteristics, either barriers or facilitators (mediating factors), to the desired health-related behaviour.¹⁹ Therefore, this methodology holds great significance in targeting interventions towards specific groups and fostering the adoption of healthy behaviours, like PA; facilitating as well the efficient prediction, and design of population-based health promotion interventions.²⁴ Over the past years, decision tree analysis has been applied in various

public health settings with different populations.²⁵ However, to our knowledge, limited research has specifically explored the application of decision trees aiming to understand mediating factors for PA among individuals with multimorbidity and at increased dementia risk.²⁶

Our primary aim is to investigate the impact of one multimodal intervention, the AgeWell study, on PA behaviour at 24-month follow-up, aiming to elucidate changes in PA after participation. Additionally, we aim to integrate decision tree analysis to understand the mediating factors influencing changes in PA at the 24-month follow-up among older adults with multimorbidity and increased dementia risk, identifying as well high-risk subgroups. These objectives will not only provide significant insights into the challenges and opportunities for promoting PA in older adults with multimorbidity and dementia risk but also offer a solid foundation for the design of future interventions and public health policies targeting this vulnerable demographic group.

Methods

Study Design

The AgeWell.de study is the first cluster-randomized, controlled trial in Germany to evaluate the effectiveness of a multicomponent intervention aimed at preventing cognitive decline in older adults with multimorbidity and at increased dementia risk.²⁷ The intervention for the experimental group encompassed a diverse range of content, including PA enhancement, which consisted of a standardized training program with home exercises (strengthening and flexibility/ balance exercises) to be performed twice a week. Additionally, individualized goals for aerobic exercise were set, targeting 3 to 5 times per week. Meanwhile, the control group received general practitioner treatment as usual and general health advice covering the components of the intervention. The primary outcome measure was cognitive function, with secondary outcomes including mental and physical health assessments. The AgeWell.de trial is registered in the German identifier: Clinical Trials Register (DRKS; trial DRKS00013555).

Participants

The study included community-dwelling patients aged between 60 and 77 years old and at increased risk of dementia, as determined by the CAIDE Dementia Risk Score (Cardiovascular Risk Factors, Aging, and Incidence of Dementia).²⁸ Participants were included in the study based on a CAIDE Dementia Risk Score of \geq 9. Furthermore, the prediction of dementia risk was determined using a sensitivity of .77 and a specificity of .63 at a cut-off point of \geq 9.²⁸ The CAIDE assessment comprises various factors, including age, education, gender, blood pressure, body mass index, total cholesterol, and physical activity. This approach ensured that individuals already diagnosed with dementia or mild cognitive impairment at the study's outset were not included, as the CAIDE Dementia Risk Score was specifically developed to assess dementia risk in individuals without clinical dementia or mild cognitive impairment at the time of evaluation. Eligible participants for AgeWell.de were screened at general practitioner (GP) practices. Exclusion criteria encompassed the diagnosis or suspicion of dementia by GP, health conditions that could impede safe participation in the intervention as determined by GP, significant loss of vision, hearing, or insufficient ability to speak or read German, severe mobility impairment and concurrent participation in any other intervention trial. For additional information regarding the diagnostic criteria and procedures employed in this study, readers are encouraged to consult the AgeWell study protocol,²⁹ which provides a comprehensive overview of the methodology and assessment procedures.

Between June 2018 and October 2019, enrolled GP practices recruited eligible participants (n = 1030) from five study sites. Among them, n = 487 participants were randomized to the intervention group, while n = 543 participants were randomized to the control group. For the intention-to-treat analysis, n = 819 (79.5%) participants (intervention/ control: n = 378/441) completed the 24-month assessment between July 2020 and January 2022. Dropout rates were n = 109 (22.4%) in the intervention group and n = 102 (18.8%) in the control group. Due to the COVID-19 pandemic and associated restrictions on personal contact, there were delays in the follow-up assessments, resulting in a mean duration of M = 25.6 months (SD: 2.8) between the baseline and follow-up interviews. The participant flow is illustrated in detail in Figure 1.

Data Collection

Data collection consisted of self-reported, structured face-toface interviews and standardized questionnaires implemented at baseline and follow-up by GPs. The different relevant study variables and measurement tools used are presented below.

Dependent Variable. We administrated a self-developed questionnaire adapted from Lippke et al.³⁰ to evaluate "regular PA". Participants were inquired about their recent participation in regular PA (≥ 2 days per week for 30 minutes), and their responses were recorded as a dichotomous variable yes or no. This reference value (regular PA = ≥ 2 days per week for 30 minutes) was chosen as a benchmark, given that regular weekly aerobic PA is linked to significant benefits on several health outcomes.¹⁵

Taking into account that one of our aims was to identify factors associated with changes in PA engagement at followup, the variable "regular PA engagement" was transformed in four PA states (changes): (1) Physically active at baseline and follow-up; (2) Physically active at baseline, became inactive at follow-up; (3) Physically inactive at baseline, became



Figure 1. Participant flowchart for random assignment in the AgeWell.de study.

regularly active at follow-up; and (4) Absence of regular PA both at baseline and follow-up.

Independent Variables. GPs provided sociodemographic (age, sex, living with a partner, education) and participant's medical conditions information. Other factors were measured through questionnaires, which included the following variables: Cognitive status (Montreal Cognitive Assessment (MoCA));³¹ Depressive symptoms (Geriatric Depression Scale (GDS));³² Performance in activities of daily living (Barthel-Index);³³ and Health-related quality of life (EQ-5D-5 L).³⁴ In addition,

participant's health behaviours, such as alcohol consumption per day (g/day), were assessed. Participants were also asked about their current smoking status, with response options of yes or no.

Furthermore, participants self-efficacy for PA was evaluated using the health-specific self-efficacy scale proposed by Schwarzer and Renner,³⁵ which assessed how confident they felt to be physically active when: they were worried and had a problem, when they felt sad, tense, tired, and busy. The available response choices were (1) Very uncertain; (2) Rather uncertain; (3) Almost certain; and (5) very certain. Social support measurements were obtained using the Lubben Social Network Scale/LSNS-6. Finally, the living area variable was examined and categorized into four classes based on population density: small urban area, with a population between 50,000 and 100,000; medium-sized urban area, with a population between 100,000 and 250,000; metropolitan urban area, with a population between 250,000 and 1.5 million; and large metropolitan urban areas, with a population above 1.5 million.³⁶

In order to identify whether there had been changes after the intervention, quantitative and qualitative independent variables were transformed into new variables. Quantitative variables (cognitive status, depression, social isolation, degree of independence, quality of life, alcohol consumption, and self-efficacy for PA) were transformed by computing the postpre score difference. As for the qualitative variables with dichotomous yes/no responses, they were transformed into variables representing four states. For instance, the variable mobility difficulties was categorized into these four states: (1) Mobility difficulties at baseline, mobility difficulties at followup; (2) Mobility difficulties at baseline, no mobility difficulties at follow-up; (3) No mobility difficulties at baseline, mobility difficulties at follow-up; and (4) No mobility difficulties at baseline, no mobility difficulties at follow-up. The same transformation was applied to the variables smoker and GPs diagnoses (type 2 diabetes, hypertension, coronary heart disease, myocardial infarction, visual and hearing difficulties). Conversely, variables such as sex, age, living with a partner, educational level, and living area remained unchanged and were utilized in their baseline form.

Statistical Data Analysis. Descriptive statistics were used for the baseline characteristics of participants. Moreover, to assess between groups differences in PA engagement outcome at 24 months follow-up, a generalized estimating equations model was performed from a repeated measures binary logistic model. Finally, a decision tree analysis was conducted to identify mediating factors associated with changes in PA state after the intervention. Using a decision tree model, the probability of different variables in determining changes in regular PA engagement was estimated. This process allowed us to prioritize those factors that could be addressed to promote changes in regular PA practice (as shown in Figure 2). Statistical tests such as Pearson's chi-square identified significant differences between subgroups, with a P-value < .05 considered significant. Analysis was done using IBM Statistical Package and Services Solutions (SPSS) version 25 (IBM Corp, Armonk, NY, USA).

Ethics. The study was approved by the responsible ethics committees of the coordinating centre (Ethics Committee of the Medical Faculty of the University of Leipzig; ethical vote number: 369/17-ek) and of all participating study sites. Participants provided written informed consent to participate.

Results

Baseline Characteristics

Characteristics of participants at baseline are reported in Table 1. At baseline, it was observed that a total of 198 participants (52.4%) in the intervention group and 232 participants (53.2%) in the control group engaged in regular physical activity at least 2 times per week for 30 minutes. There were no significant differences between the two groups in terms of regular physical activity participation at baseline (P = .529). Participants who dropped out during the intervention did not differ from those who completed the follow-up assessment in terms of sex (P = .353), age (P = .443), or group allocation (P = .153). Those who dropped out more often before the follow-up assessment had lower education levels than participants who completed the follow-up (P < .001).

Between-Group Differences in PA Engagement at 24 Months Follow-Up

A generalized estimating equations model was conducted based on a repeated measures binary logistic model, in which it was identified that there were no significant differences between the control and the experimental group. Results showed a *P*-value equal to .674.

Although there were no significant differences between groups, some changes in PA behaviour were found. Figure 3 provides a description of the four different PA states (changes) that resulted from baseline to follow-up between-groups. Particularly, while 52.7% of the participants in the intervention group became physically active after the intervention, only 47.3% of the control group were able to make this transition at follow-up. Also, a greater absence of PA can be observed in the control group after the intervention (56.4%), in contrast to the intervention group (43.6%).

Outcomes of Decision Trees Analysis

Mediating Factors for Changes in PA States. A decision tree analysis was conducted to recognise mediating factors associated with previous changes in PA states between baseline and follow-up. We employed decision trees as a method to dissect the intricate relationships between various factors and their impact on PA bahaviour. Decision trees serve as a visual representation to understand these relationships more intuitively. The process begins by examining the entirety of the dataset, similar to starting at the root of a tree. Subsequently, the algorithm divides the dataset into smaller subsets based on the most influential factors affecting PA. This segmentation continues recursively, forming a tree-like structure. Each terminal node of the tree represents a distinct subgroup of individuals with similar characteristics. By exploring the branches from the root to the terminal nodes, we can identify the primary mediating factors influencing PA patterns.



Figure 2. Framework analysis employed for recognising factors mediating regular PA changes.

According to our analyisis, illustrated in Figure 3, the model identified that the most influential variable impacting changes in PA was the difference in self-efficacy scores (P < .001). This finding suggests that variations in self-efficacy scores between baseline and follow-up influenced the likelihood of engaging in PA. Additionally, as illustrated in Figure 4, the model categorized the difference in self-efficacy scores into five distinct groups: Node (1) score difference equal or less than -6 points; Node (2) score difference between -1 and 0 points; Node (4) score difference between 0 and 5 points; Node (5) score difference greater than 5 points. Notably, positive scores indicated an increase in self-efficacy, whereas negative scores indicated a decrease in self-efficacy at follow-up.

Nodes 1, 2, and 3 indicated a decrease in the probability of transitioning from physical inactivity at baseline to becoming physically active at follow-up among individuals with a negative difference in self-efficacy scores ranging from \leq -6 to 0 points (as indicated by the blue square in Figure 4). For instance, in node 1, this probability dropped from 11.9% to 8.3%; in node 2, the probability went from 11.9% to 8.6%; and node 3 declined from 11.9% to 7.4%. On the contrary, nodes 4 and 5 showed that individuals who increased scores from 0 to \geq 5 points on the self-efficacy scale raised their likelihood of transitioning to a state of regular PA during follow-up after being inactive at baseline. For example, the probability at node 4 increased from 11.9% to 14.5% and in node 5 raised to 27.3%.

Moreover, the findings in nodes 6 and 7 revealed a significant association between gender and a negative difference of ≤ 6 points in self-efficacy scores (P = .0029). For instance, men experiencing a decline in their self-efficacy scores by \leq -6 points increased their probability of absence of regular PA both at baseline and follow-up (PA state indicated by red square in Figure 4). This probability increased from 52.7% to 63.3%. On the contrary, despite having the same negative difference in their self-efficacy scores, women decreased this probability from 52.7% to 40.5%.

Furthermore, node 8 highlights that individuals with diabetes and a self-efficacy score difference ranging from -6 to 1 point showed an increased likelihood of absence of regular PA, both at baseline and at later follow-up (PA state indicated by red square in Figure 4). This probability increased from 27.6% to 44.4%. In comparison, node 9 indicated that those not diagnosed with this disease reduced this probability from 27.6% to 20.4%.

Finally, nodes 10 and 11 showed a significant relationship between an increase in self-efficacy scores (between 0 and 5 points) and the condition of the participants, either belonging to the control or intervention groups (P =.009). In Node 10, participants within the intervention group who experienced a self-efficacy score increase of 0 to 5 points at follow-up were observed to have an augmented probability of transitioning from physical inactivity at baseline to regular activity at follow-up (as indicated by the blue square in Figure 3). This probability increased from 14.5% to 22.7%. Conversely, in node 11, participants from

Variables	Intervention (n = 378)	Control (n = 441)
Dependent variable		
Regular PA at least 2 times/week for 30 min (yes), n (%)	198 (52.4)	232 (53.2)
Independent variables		
Individual factors		
Sociodemographic characteristics		
Age in years, mean (SD)	69.1 (4.9)	69.0 (4.9)
Sex, female (yes), n (%)	199 (52.7)	234 (53.1)
Living with a partner (yes), n (%)	247 (65.3	285 (64.6)
Education		
Low (yes), n (%)	98 (25.9)	83 (18.8)
Intermediate (yes), n (%)	196 (51.9)	238 (54.0)
High (yes), n (%)	84 (22.2)	120 (27.2)
Morbidities profile of older adults at increased dementia risk accordir	ng to GPs diagnosis	()
Diabetes mellitus type 2 (yes), n (%)	136 (36.0)	175 (39.7)
Hypertension (yes), n (%)	332 (87.8)	375 (85.0)
Coronary heart disease (yes), n (%)	66 (17.5)	70 (15.9)
Myocardial infarction (yes), n (%)	23 (6.1)	25 (5.7)
Visual difficulties (yes), n (%)	14 (3.7)	32 (7.3)
Hearing difficulties (yes), n (%)	4 (1.1)	5 (I.I)
Mobility difficulties (yes), n (%)	(.3)	2 (.5)
Cognitive status (MoCA sum score points), mean (SD)	24.47 (2.9)	24.75 (3.0)
Activities of daily living (Barthel index) points, mean (SD)	99.64 (2.3)	99.64 (2.5)
Depression (GDS scale) points, mean (SD)	1.48 (1.9)	1.48 (1.9)
Health-related quality of life (EQ-5D-5 L VAS), mean (SD)	78.27 (15.1)	75.62 (16.0)
Alcohol consumption (g/day)	14.21 (24.7)	15.28 (23.8)
Smoker (ves), n (%)	46 (12.2)	43 (9.8)
Self-efficacy for PA		
I feel confident to be physically active when		
I have concerns and problems (yes), n (%)	298 (78.8)	310 (70.6)
When I am feeling down (yes), n (%)	265 (70.1)	278 (63.4)
When I feel tense (ves), n (%)	279 (73.8)	300 (68.3)
When I am tired (yes), n (%)	207 (54.8)	184 (41.9)
I am busy (ves), n (%)	175 (46.3)	173 (39.5)
Interpersonal factors		
Social support (Lubben social network scale) points, mean (SD)	21.19 (6.4)	20.88 (6.7)
Environmental factors		
Living area		
Small urban area, (ves), n (%)	70 (18.5)	96 (21.8)
Medium-sized urban area, (ves), n (%)	159 (42.1)	159 (36.1)
Metropolitan urban area, (yes), n (%)	[]7 (31.0)	120 (27.2)
Large metropolitan urban area, (yes), n (%)	32 (8.5)	66 (15.0)

Note. PA = physical activity, SD = standard deviation, GPs = general practitioners, MoCA = Montreal-Cognitive- Assessment, GDS = Geriatric Depression Scale.

the control group demonstrated a decrease in this probability from 14.5% to 8.4%.

In summary, the results of the decision tree analysis reveal that the difference in self-efficacy scores was the most influential variable in changes in physical activity states between baseline and follow-up. The probability of engaging in physical activity varied significantly based on the difference in self-efficacy scores, indicating an increased likelihood of regular physical activity participation with higher self-efficacy scores. Furthermore, significant associations were observed between gender and the presence of diabetes with changes in self-efficacy and physical activity participation. Additionally, a significant relationship was identified between the increase in self-efficacy scores and the participants' condition, with a notable increase in the probability of transitioning to regular physical activity in the experimental group compared to the



Figure 3. Between-groups changes in PA states from baseline to follow-up.



Figure 4. Decision tree analysis: Mediating factors associated with PA changes at follow-up.

control group. These findings underscore the importance of self-efficacy in promoting physical activity and suggest that interventions aimed at improving self-efficacy may be effective in fostering more active behaviors in the studied population.

Mediating Factors for PA Engagement at Follow-Up. Additionally, we examined potential mediating factors influencing regular PA at follow-up. Thus, a second decision tree analysis was conducted, including just the dichotomous dependent variable "regular PA engagement ≥ 2 days/week for at least 30min"; yes/no at follow-up. The results of this analysis, as shown in Figure 5, reaffirmed self-efficacy as a significant determinant for regular PA engagement (P = .019).

Results from this tree analysis revealed (see Figure 5) that a negative difference between baseline and follow-up of \leq -4.50 points on the self-efficacy scale decreases the probability of regular PA at follow-up. This probability went from 47.5% to 29.5%. In contrast, when participants increased score differences by more than -4.5 points on the self-efficacy scale, their chances of doing PA regularly at follow-up increased. The probability rose from 47.5% to 50.4%.

In the same way, individuals who scored differences above -4.5 points on the self-efficacy scale in the follow-up and reported a high education level demonstrated an increased probability of engaging regularly in PA. This percentage went from 50.4% to 64.4%. Nonetheless, in the case of individuals with a high level of education, if they scored a difference of \leq -5.5 points in their social support scores, their likelihood of



Figure 5. Decision tree analysis for regular PA engagement at follow-up.

engaging in PA regularly in the follow-up period decreased. The probability went from 64.4% to 59.2%.

In contrast, those with an intermediate to low educational level exhibited a reduction in their probability of engaging regularly in PA at follow-up. This went from 50.4% to 45.5%. Within this group of participants who scored differences above -4.5 points on the self-efficacy scale and had a low level of education, alcohol consumption emerged as a significant variable (P = .006). Those who increased their consumption (>2.03 g per day) at follow-up were less likely to not engage in PA regularly after the intervention. This probability was reduced from 54.5% to 46.8%. On the other hand, those who decreased their alcohol consumption by -7.6 g increased their probability of doing PA regularly at follow-up. This probability changed from 44.0% to 55.9%. On the other hand, those who increased their consumption by more than -7.6 g per day decreased their likelihood of PA from 44.0% to 41.8%.

Discussion

Main Findings

In the initial analysis, the present study did not find any significant effect on PA engagement during the 24-month follow-up period. However, looking in more detail through decision trees analysis, the likelihood of regularly engaging in PA increased during the follow-up period within specific subgroups. Particularly, individuals with increased selfefficacy at follow-up raised their probability of engaging regularly in PA after the intervention. Thus, self-efficacy was identified as the most important mediating factor associated with the likelihood of changes in PA participation at followup. In addition, other characteristics of participants mediated the probability of regular PA at follow-up associated with selfefficacy, including gender, presence of diabetes, being in the treatment group, educational level, social isolation, and alcohol consumption. A more detailed discussion of these subgroups is presented below.

Self-Efficacy

The decision tree analysis identified among all the factors examined that the most important mediating factor associated with changes in regular PA participation at 24-month followup was the change in self-efficacy scores. These findings are consistent with those reported in previous literature.³⁷ For instance, self-efficacy was related to PA engagement and adherence in asymptomatic and older adults with chronic diseases, including recovered individuals of cardiovascular disease and survival in chronic obstructive pulmonary disease.^{37,38} Moreover, according to 2- to 5-year follow-up study in which PA was promoted, self-efficacy has been shown to be crucial in the long-term maintenance of PA.³⁹ Of note, the relationship between PA and self-efficacy is not unidirectional and can be influenced by other factors, such as physical and mental health, social environment, and individual limitations.³⁷ However, in general, it has been found that there is a positive association between self-efficacy and PA in older adults.

Self-efficacy refers to a person's belief in his or her ability to achieve certain tasks or goals.⁴⁰ In the context of PA, selfefficacy relates to the confidence a person has in his or her ability to participate in and maintain regular PA.⁴¹ Thus, selfefficacy may play a crucial role in exercise behaviours as older adults with higher self-efficacy perceive that they can overcome the barriers and challenges associated with PA, increasing their motivation and likelihood to participate in it.⁴¹ According to Bandura's theory, confidence in personal ability influences choices and effort devoted to PA behaviour.⁴⁰

In this way, it is necessary to address psychological key components, such as self-efficacy, to design effective programs promoting PA engagement and adherence.⁴¹ To our knowledge, there is a limited amount of research on PA interventions for patients with multimorbidity and at increased dementia risk, focusing on improving self-efficacy in PA. Thus, self-efficacy enhancement strategies may be useful in interventions to increase PA engagement and adherence among this particular population.

Self-Efficacy and Gender

Another variable that mediated regular PA engagement in our study was gender. The results of the decision tree model showed that being male was a risk factor for physical inactivity as long as there was a decrease in self-efficacy. In particular, it was found that males with a decrease in self-efficacy scores were more likely to not participate in regular PA, both at baseline and at subsequent follow-up. These results are consistent with previous literature,⁴² particularly coinciding with differences between PA determinants according to gender. One study reported that intention and habit are key to short- and long-term factors influencing PA in both genders.⁴³ However, differences were found in the role of self-efficacy, emphasizing that attitude towards PA was important for men while planning was crucial for women.⁴³ However, it is important to acknowledge the complexity of the relationship between attitude and planning in the context of PA. While our initial discussion highlighted the importance of attitude for men and planning for women, it is recognized that these factors may not operate independently.⁴³ Indeed, positive attitudes towards PA may enhance motivation for planning, suggesting a more complex relationship between attitudes and planning across genders.⁴³ In another study,⁴⁴ investigating gender differences in psychosocial factors related to PA during a lifestyle change intervention, it was found that men received more social support for PA, but women increased their exercise plans more. While there were differences in the importance of planning between genders, these were not consistent across all analyses. Self-efficacy was a strong predictor of behaviour change, and pre-intervention social support was important for both genders. Women showed greater use and effects of action planning, possibly linked to their social roles and responsibilities. Overall, the findings suggest that men and women may not differ in socio-cognitive variables generally, but rather in those related to their social roles. Interventions could benefit by supporting women more in behaviour change planning and by facilitating their ability to activate social support sources. Furthermore, existing literature has indicated gender differences in psychological resilience among older adults.⁴⁵ Given that women tend to report lower levels of psychological resilience compared to men, it is conceivable that this disparity could also contribute to differences in self-efficacy.46 Lower resilience levels among older women might influence their confidence in engaging in PA and planning their exercise routines effectively.47 Therefore, attitudes towards PA and psychological resilience could certainly be underlying factors contributing to the observed gender differences in self-efficacy.⁴⁸ Future research could explore these aspects further to provide a more comprehensive understanding of the mechanisms driving gender disparities in self-efficacy among older adults. These gender differences have implications for the development of future gender-specific PA interventions, as also stated in previous studies.43

Self-Efficacy and Type 2 Diabetes

It was also found that a decrease in self-efficacy in combination with a diagnosis of type 2 diabetes was a risk factor for physical inactivity. In other words, if participants had this diagnosis before or after the intervention and a decrease in their self-efficacy scores, they were more likely to have no change in their physical inactivity behaviour. Previous studies also show that the diagnosis of type 2 diabetes was a risk factor for inactivity.⁴⁹ Therefore, developing tailored guidelines for PA and self-efficacy enhancement aimed at this population is also essential.

Self-Efficacy in the Experimental Group vs. Control Group

Increases in self-efficacy at follow-up and intervention group participants experiencing an increase in self-efficacy between baseline and follow-up were more likely to present a higher probability of transitioning from being physically inactive at baseline to physically active in the follow-up, unlike the control group. Thus, as has also been shown in other studies,⁵⁰ multicomponent programs should aim at integrating PA, cognitive and social activities, providing knowledge, tools, and skills that may enable participants to increase self-efficacy and feel more confident in engaging in regular, long-term PA.

Self-Efficacy, Educational Level, Social Support, and Alcohol Consumption

Additionally, high educational level accompanied by increased self-efficacy at follow-up increased the likelihood of regular PA in the follow-up. Comparing these findings with recent literature, various studies showed a positive correlation between educational level and PA participation in older adults.^{51,52} A person's education can influence his or her behaviour and attitudes toward healthy lifestyle behaviours.^{53,54} Older adults with higher levels of education tend to have a better understanding of the health benefits of PA and a greater awareness of how to maintain a healthy lifestyle.⁵³ In addition, education can provide them with the skills and knowledge needed to participate in PA safely and effectively.⁵³ These findings highlight the importance of education and self-efficacy as facilitating factors to promote an active lifestyle in older adults with multimorbidity and at increased dementia risk.

It is worth noting that for this population with increased self-efficacy and a high level of education, a decrease in social support at follow-up was a risk factor for not engaging in regular PA. Thus, these results highlight that social support was a relevant mediator for PA in people with higher education and multimorbidity. We did not find studies showing direct correlations between high educational levels, decreased social support and PA engagement. In general, studies reported that a high level of education was associated positively with social support as it offers greater opportunities for social interaction and access to broader social networks.⁵⁵ However, each population group is different. Perhaps there are certain additional key aspects to consider when individuals are multimorbid and at increased dementia risk. For instance, although individuals may have a higher level of education and selfefficacy when they suffer from multimorbidity, they may experience changes in their social environment and support. which could affect their probability of engaging in PA. Hence, social support might be also an essential factor to consider when planning PA promotion strategies for persons with multimorbidity and at increased dementia risk.

Finally, low educational level and alcohol consumption were also risk factors, as these factors reduced the likelihood of regular PA in follow-up. These findings are in accordance with the literature, as it has been found that low educational levels and alcohol consumption are associated with low PA levels.⁵⁶ This may be due to unfavourable socioeconomic factors, social and cultural norms that normalize alcohol consumption and hinder access to resources, knowledge, and skills needed to initiate and maintain PA behaviour.⁵⁷ Programs aimed at increasing self-efficacy among low-educated older adults should consider the challenges and limitations inherent to this population sub-group. Such programs could incorporate components and strategies that guarantee access to resources, support, knowledge, and skill building.

Strengths and Limitations

One of the strengths of this study lies in the utilization of decision tree analysis. This approach facilitated examine complex interactions among mediating factors for PA engagement during the follow-up period and identified high-risk subgroups. The classification of the population based on these findings will contribute to the effective targeting and design of future interventions targeted at high-risk groups for dementia. The study has recognized various limitations, highlighting the need to interpret the findings carefully. Firstly, larger sample sizes are necessary to ensure more robust and reliable results. Furthermore, the exact amount of PA undertaken by individuals is unknown, as only the receipt of the multimodal intervention is documented. Lastly, the sample was rather selective - based on the inclusion criteria alone, an increased number of participants were probably physically inactive (physical inactivity is also a component of the CAIDE score, and criteria such as body-mass-index (BMI) > 30, which also correlates strongly with inactivity). Therefore, the results can only be generalized to a limited extent.

Conclusion

The study sample did not exhibit any significant impact on PA effect following the intervention. However, certain subgroups showed an increased likelihood of adopting regular PA during the follow-up period. Notably, participants who experienced an improvement in their self-efficacy scores at follow-up had a higher probability of enhancing their regular PA, mainly if they were initially physically inactive. These observations suggest that intervention programs successfully enhancing self-efficacy are more likely to promote regular PA. Therefore, self-efficacy remains a crucial aspect to consider and evaluate in future intervention studies targeting older adults with multimorbidity and at increased dementia risk. Efforts should be focused on individualizing interventions, particularly for men, individuals with type 2 diabetes, those with low social support, those with intermediate or low education, and those with higher alcohol consumption.

Appendix

Abbreviations

BMI	Body-mass-index
CAIDE	Cardiovascular Risk Factors, Aging, and Incidence
	of Dementia
GDS	Geriatric Depression Scale
GP	General practitioner
LSNS	Lubben Social Network
MoCA	Montreal Cognitive Assessment
PA	Physical activity
SPSS	Statistical Package and Services Solutions
NCDs	Non-communicable diseases

WHO World Health Organization.

Acknowledgments

Members of the AgeWell.de study group: Principal Investigator: Steffi G. Riedel-Heller; Co-Principal Investigators: Wolfgang Hoffmann, Jochen Gensichen, Walter E. Haefeli, Hanna Kaduszkiewicz, Hans-Helmut König, Thomas Frese, David Czock, Jochen René Thyrian; Members of the study team: Franziska Berg, Andrea Bischhoff, Christian Brettschneider; Mandy Claus, Juliane Döhring, Alexander Eßer, Corinna Gräble, Stephanie Hingst, Caroline Jung-Sievers, Kerstin Klauer-Tiedtke, Kerstin Krebs-Hein, Flora Kühne, Sebastian Lange, Paula Liegert, Dagmar Lochmann, Tobias Luck, Silke Mamone, Andreas Meid, Michael Metzner, Lydia Neubert, Anke Oey, Susanne Röhr, Franziska-Antonia Zora Samos, Karin Schumacher, Theresa Terstegen, Sandy Thieme, Svetlana Usacheva, Lars Wamsiedler, Tanja Wehran, Marina Weißenborn, Ines Winkler, Isabel Zöllinger, Andrea Zülke, Ina Zwingmann. Additionally, we would like to extend our acknowledgments to Julian Tamayo for his statistical guidance in the decision trees analysis. His expertise and insights greatly contributed to the rigor and accuracy of our study.

Author Contributions

Conceptualization, J.G., W.H., H.K., H.-H.K., J.R.T., B.W., and S.G.R.-H.; Data curation, M.I.C., and B.W.; Formal analysis, M.I.C.; Funding acquisition, J.G., W.H., H.K., H.-H.K., J.R.T., B.W., and S.G.R.-H.; Investigation, M.I.C., A.Z., M.L., C.B., A.B., E.S.K., and J.D.; Methodology, M.I.C. and J.R.T.; Project administration, W.H., H.K., H.-H.K., J.R.T., B.W., and S.G.R.-H.; Resources, J.G., W.H., H.K., H.-H.K., J.R.T., and S.G.R.-H.; Supervision, A.Z., M.L., T.F., J.G., W.H., H.K., J.R.T., B.W., and S.G.R.-H.; Validation, W.H., H.K., T.F., J.G., J.R.T., B.W., and S.G.R.-H.; Visualization, M.I.C. Writing—original draft, M.I.C.; Writing—review and editing, E.S.K., A.B., M.W., J.G., M.L., A.Z., W.H., H.K., C.E., R.P.K., J.D., C.B., H.-H.K., B.W., J.R.T., S.G.R.-H. All authors have read and agreed to the published version of the manuscript.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This publication is part of the study "AgeWell.de – a multi-centric clusterrandomized controlled prevention trial in primary care" and was funded by the German Federal Ministry for Education and Research (BMBF; grants: 01GL1704A, 01GL1704B, 01GL1704C, 01GL1704D, 01GL1704E, 01GL1704F).

Ethical Statement

Ethical Approval

The study was conducted according to the guidelines of the Declaration of Helsinki, and approved by the Ethics Committee of the Medical Faculty of the University of Leipzig (ethical vote number: 369/17-ek; 12 October 2017).

Informed Consent

Informed consent was obtained from all participants involved in the study.

ORCID iDs

Maria Isabel Cardona b https://orcid.org/0000-0003-2614-4670 Christian Brettschneider b https://orcid.org/0000-0002-5280-1075 Wolfgang Hoffmann b https://orcid.org/0000-0002-6359-8797

Data Availability Statement

The data presented in this study are available on request from the corresponding author.

References

- Prince MJ, Wimo A, Guerchet MM, Ali GC, Wu Y-T, Prina M. World Alzheimer Report 2015-The Global Impact of Dementia: An Analysis of Prevalence, Incidence, Cost and Trends. London: Alzheimer's Disease International; 2015.
- Estimation of the global prevalence of dementia in 2019 and forecasted prevalence in 2050: an analysis for the Global Burden of Disease Study 2019. *Lancet Public Health*. 2022;7(2): e105-e125. doi:10.1016/s2468-2667(21)00249-8
- 3. *Dementia*. Geneva, Switzerland: World Health Organization; 2022.
- Livingston G, Huntley J, Sommerlad A, et al. Dementia prevention, intervention, and care: 2020 report of the Lancet Commission. *Lancet*. 2020;396(10248):413-446. doi:10.1016/ s0140-6736(20)30367-6
- World Health Organization. Global Action Plan on the Public Health Response to Dementia 2017–2025. Geneva, Switzerland: World Health Organization; 2017.
- Dukelow T, Lawrence EG, Jacobson L, et al. Modifiable risk factors for dementia, and awareness of brain health behaviors: results from the Five Lives Brain Health Ireland Survey (FLBHIS). *Front Psychol.* 2022;13:1070259. doi:10.3389/ fpsyg.2022.1070259
- Grande G, Marengoni A, Vetrano DL, et al. Multimorbidity burden and dementia risk in older adults: the role of inflammation and genetics. *Alzheimers Dement*. 2021;17(5):768-776. doi:10.1002/alz.12237
- Johnston MC, Crilly M, Black C, Prescott GJ, Mercer SW. Defining and measuring multimorbidity: a systematic review of systematic reviews. *Eur J Publ Health*. 2019;29(1):182-189. doi:10.1093/eurpub/cky098
- Ngandu T, Lehtisalo J, Solomon A, et al. A 2 year multidomain intervention of diet, exercise, cognitive training, and vascular risk monitoring versus control to prevent cognitive decline in atrisk elderly people (FINGER): a randomised controlled trial. *Lancet.* 2015;385(9984):2255-2263. doi:10.1016/s0140-6736(15)60461-5

- Andrieu S, Guyonnet S, Coley N, et al. Effect of long-term omega 3 polyunsaturated fatty acid supplementation with or without multidomain intervention on cognitive function in elderly adults with memory complaints (MAPT): a randomised, placebo-controlled trial. *Lancet Neurol*. 2017;16(5):377-389. doi:10.1016/s1474-4422(17)30040-6
- Moll van Charante EP, Richard E, Eurelings LS, et al. Effectiveness of a 6-year multidomain vascular care intervention to prevent dementia (preDIVA): a cluster-randomised controlled trial. *Lancet*. 2016;388(10046):797-805. doi:10.1016/s0140-6736(16)30950-3
- Cardona MI, Afi A, Lakicevic N, Thyrian JR. Physical activity interventions and their effects on cognitive function in people with dementia: a systematic review and meta-analysis. *Int J Environ Res Publ Health*. 2021;18(16):8753. doi:10.3390/ ijerph18168753
- Langhammer B, Bergland A, Rydwik E. The importance of physical activity exercise among older people. *BioMed Res Int.* 2018;2018:7856823. doi:10.1155/2018/7856823
- Martinez-Gomez D, Guallar-Castillon P, Garcia-Esquinas E, Bandinelli S, Rodríguez-Artalejo F. Physical activity and the effect of multimorbidity on all-cause mortality in older adults. *Mayo Clin Proc.* 2017;92(3):376-382. doi:10.1016/j.mayocp. 2016.12.004
- 15. *Global Recommendations on Physical Activity for Health.* Geneva, Switzerland: World Health Organization; 2010.
- Davis MG, Fox KR. Physical activity patterns assessed by accelerometry in older people. *Eur J Appl Physiol*. 2007;100(5): 581-589. doi:10.1007/s00421-006-0320-8
- Gomes M, Figueiredo D, Teixeira L, et al. Physical inactivity among older adults across Europe based on the SHARE database. *Age Ageing*. 2017;46(1):71-77. doi:10.1093/ageing/ afw165
- Daten und Fakten: Ergebnisse der Studie. Gesundheit in Deutschland Aktuell 2012. Berlin: Robert Koch-Institute; 2014.
- Salman A, Sellami M. Do older adults with multimorbidity meet the recommended levels of physical activity? An analysis of scottish health survey. *Int J Environ Res Publ Health*. 2019; 16(19):3748. doi:10.3390/ijerph16193748
- Jäger M, Lindhardt MC, Pedersen JR, et al. Putting the pieces together: a qualitative study exploring perspectives on self-management and exercise behavior among people living with multimorbidity, healthcare professionals, relatives, and patient advocates. J Multimorb Comorb. 2022;12: 26335565221100172. doi:10.1177/26335565221100172
- Cardona MI, Weißenborn M, Zöllinger I, et al. Physical Activity Determinants in Older German Adults at Increased Dementia Risk with Multimorbidity: Baseline Results of the AgeWell.de Study. *Int J Environ Res Publ Health*. 2022;19(6):3164. doi:10. 3390/ijerph19063164
- McLeroy KR, Bibeau D, Steckler A, Glanz K. An ecological perspective on health promotion programs. *Health Educ Q*. 1988;15(4):351-377. doi:10.1177/109019818801500401
- 23. Bricca A, Harris LK, Jäger M, Smith SM, Juhl CB, Skou ST. Benefits and harms of exercise therapy in people with

multimorbidity: a systematic review and meta-analysis of randomised controlled trials. *Ageing Res Rev.* 2020;63:101166. doi:10.1016/j.arr.2020.101166

- Lemon SC, Roy J, Clark MA, Friedmann PD, Rakowski W. Classification and regression tree analysis in public health: methodological review and comparison with logistic regression. *Ann Behav Med.* 2003;26(3):172-181. doi:10.1207/ s15324796abm2603 02
- Lemon SC, Roy J, Clark MA, Friedmann PD, Rakowski W. Classification and regression tree analysis in public health: methodological review and comparison with logistic regression. *Ann Behav Med.* 2003;26(3):172-181. doi:10.1207/ s15324796abm2603_02
- Khunti K, Highton PJ, Waheed G, et al. Promoting physical activity with self-management support for those with multi-morbidity: a randomised controlled trial. *Br J Gen Pract.* 2021; 71(713):e921-e930. doi:10.3399/bjgp.2021.0172
- 27. Zülke A, Pabst A, Luppa M, et al. A multidomain intervention against cognitive decline in an at-risk-population in Germany results from the cluster-randomized AgeWell.de-trial. *Alz-heimers Dement.* 2024;20:615-628.
- Kivipelto M, Ngandu T, Laatikainen T, Winblad B, Soininen H, Tuomilehto J. Risk score for the prediction of dementia risk in 20 years among middle aged people: a longitudinal, populationbased study. *Lancet Neurol.* 2006;5(9):735-741. doi:10.1016/ s1474-4422(06)70537-3
- Zülke A, Luck T, Pabst A, et al. AgeWell.de study protocol of a pragmatic multi-center cluster-randomized controlled prevention trial against cognitive decline in older primary care patients. *BMC Geriatr.* 2019;19(1):203. doi:10.1186/s12877-019-1212-1
- Lippke S, Fleig L, Pomp S, Schwarzer R. Validity of a stage algorithm for physical activity in participants recruited from orthopedic and cardiac rehabilitation clinics. *Rehabil Psychol*. 2010;55(4):398-408. doi:10.1037/a0021563
- Nasreddine ZS, Phillips NA, Bédirian V, et al. The Montreal Cognitive Assessment, MoCA: a brief screening tool for mild cognitive impairment. *J Am Geriatr Soc.* 2005;53(4):695-699. doi:10.1111/j.1532-5415.2005.53221.x
- Gauggel S, Birkner B. Validität und Reliabilität einer deutschen Version der Geriatrischen Depressionsskala (GDS). Z Klin Psychol Psychiatr Psychother. 1999;28(1):18-27. doi:10.1026// 0084-5345.28.1.18
- 33. Mahoney FI, Barthel DW. Functional evaluation: the barthel index. *Md State Med J.* 1965;14:61-65.
- Lin DY, Cheok TS, Samson AJ, et al. A longitudinal validation of the EQ-5D-5L and EQ-VAS stand-alone component utilising the Oxford Hip Score in the Australian hip arthroplasty population. J Patient Rep Outcomes. 2022;6(1):71. doi:10.1186/ s41687-022-00482-7
- Schwarzer R, Renner B. Social-cognitive predictors of health behavior: action self-efficacy and coping self-efficacy. *Health Psychol.* 2000;19(5):487-495.
- Dijkstra L, Poelman H, Veneri P. *The EU-OECD Definition of a Functional Urban Area*. Paris: OECD iLibrary; 2019. doi:10. 1787/d58cb34d-en

- Collado-Mateo D, Lavín-Pérez AM, Peñacoba C, et al. Key factors associated with adherence to physical exercise in patients with chronic diseases and older adults: an umbrella review. *Int J Environ Res Publ Health*. 2021;18(4):2023. doi:10.3390/ ijerph18042023
- Kaplan RM, Atkins CJ, Reinsch S. Specific efficacy expectations mediate exercise compliance in patients with COPD. *Health Psychol.* 1984;3(3):223-242. doi:10.1037//0278-6133.3. 3.223
- McAuley E, Morris KS, Motl RW, Hu L, Konopack JF, Elavsky S. Long-term follow-up of physical activity behavior in older adults. *Health Psychol.* 2007;26(3):375-380.
- Bandura A. Social Foundations of Thought and Action: A Social Cognitive Theory. Hoboken, NJ: Prentice Hall; 1986.
- Lee LL, Arthur A, Avis M. Using self-efficacy theory to develop interventions that help older people overcome psychological barriers to physical activity: a discussion paper. *Int J Nurs Stud.* 2008;45(11):1690-1699. doi:10.1016/j.ijnurstu.2008.02.012
- Godin G, Shephard RJ. Gender differences in perceived physical self-efficacy among older individuals. *Percept Mot Skills*. 1985; 60(2):599-602. doi:10.2466/pms.1985.60.2.599
- Tummers S, Hommersom A, Lechner L, Bemelmans R, Bolman CAW. Determinants of physical activity behaviour change in (online) interventions, and gender-specific differences: a Bayesian network model. *Int J Behav Nutr Phys Activ.* 2022; 19(1):155. doi:10.1186/s12966-022-01381-2
- Hankonen N, Absetz P, Ghisletta P, Renner B, Uutela A. Gender differences in social cognitive determinants of exercise adoption. *Psychol Health*. 2010;25(1):55-69. doi:10.1080/ 08870440902736972
- Böell JE, Silva DM, Hegadoren KM. Sociodemographic factors and health conditions associated with the resilience of people with chronic diseases: a cross sectional study. *Rev Lat Am Enfermagem.* 2016;24:e2786. doi:10.1590/1518-8345.1205. 2786
- Whitesides L, Lynn J. The psychosocial effects of frailty on women. *Clin Geriatr Med.* 2021;37(4):543-552. doi:10.1016/j. cger.2021.05.003
- Rodrigues FR, Tavares D. Resilience in elderly people: factors associated with sociodemographic and health conditions. *Rev Bras Enferm.* 2021;74(suppl 2):e20200171. doi:10.1590/0034-7167-2020-0171
- Górska S, Singh Roy A, Whitehall L, Irvine Fitzpatrick L, Duffy N, Forsyth K. A systematic review and correlational metaanalysis of factors associated with resilience of normally aging, community-living older adults. *Gerontol.* 2022;62(9): e520-e533. doi:10.1093/geront/gnab110
- Mahmudiono T, Setyaningtyas SW, Rachmah Q, et al. Selfefficacy in physical activity and glycemic control among older adults with diabetes in Jagir Subdistrict, Surabaya, Indonesia. *Heliyon*. 2021;7(7):e07578. doi:10.1016/j.heliyon.2021.e07578
- Streber A, Abu-Omar K, Hentschke C, Rütten A. A multicenter controlled study for dementia prevention through physical, cognitive and social activities - GESTALT-kompakt. *Clin Interv Aging*. 2017;12:2109-2121. doi:10.2147/cia.S14116

- Guiney H, Keall M, Machado L. Physical activity prevalence and correlates among New Zealand older adults. *J Aging Phys Activ.* 2018;26(1):75-83. doi:10.1123/japa.2016-0393
- Liu YB, Liu L, Li YF, Chen YL. Relationship between health literacy, health-related behaviors and health status: a survey of elderly Chinese. *Int J Environ Res Publ Health*. 2015;12(8): 9714-9725. doi:10.3390/ijerph120809714
- Korkmaz Aslan G, Kartal A, Özen Çınar İ, Koştu N. The relationship between attitudes toward aging and health-promoting behaviours in older adults. *Int J Nurs Pract*. 2017;23(6):e12594. doi:10.1111/ijn.12594
- 54. Kulakçı H, Kuzlu Ayyıldız T, Emiroğlu O, Köroğlu E. Evaluation of self-efficacy perceptions and healthy lifestyle

behaviors of the elderly living in the nursing home. *Dokuz Eylul Univ Hemsirelik Yuksekokulu Electron J.* 2012;5(2): 53-64.

- Lee S-YD, Arozullah AM, Cho YI, Crittenden K, Vicencio D. Health literacy, social support, and health status among older adults. *Educ Gerontol.* 2009;35(3):191-201. doi:10.1080/ 03601270802466629
- Boing AF, Subramanian SV, Boing AC. Association between area-level education and the co-occurrence of behavior-related risk factors: a multilevel analysis. *Rev Bras Epidemiol*. 2019;22: e190052. doi:10.1590/1980-549720190052
- Collins SE. Associations between socioeconomic factors and alcohol outcomes. *Alcohol Res.* 2016;38(1):83-94.